Dropout from Cognitive Behavioural Treatment in a Case of Bulimia Nervosa: the Role of the Therapeutic Alliance

Diego Sarracino^{1^{IX}}, Alice Garavaglia¹, Emanuela Gritti¹, Laura Parolin¹, & Marco Innamorati²

Abstract. Despite the refinement of the cognitive treatment for eating disorders, relatively high dropout rates represent a major problem for therapists and researchers. This study investigated the case of a patient with a diagnosis of bulimia nervosa, who dropped out of outpatient CBT after 28 weekly sessions. In addition to standard clinical outcome assessment, we examined how patient's psychological functioning and therapeutic alliance changed across sessions by applying observer-rating scales to the therapy transcripts. Although the patient reported some improvement at the six-month retest, observer ratings showed persistence of impaired functioning and frequent ruptures in the patient-therapist relationship throughout the treatment. We concluded that a thorough examination of the therapy process might help to understand the factors that lead to premature treatment termination.

Keywords: eating disorders, dropout, multistep cognitive behavior therapy, process research, therapeutic alliance

In an effort to accelerate dynamic psychotherapy, Davanloo began in the 1960s to use more active approaches to identify and handle resistances. He initially worked with low and moderate resistant patients who could be put in touch with avoided unconscious feelings through focal interviewing, active use of interpretation of defenses and interruption of defenses that came into play (Davanloo, 1980).

Many studies have been conducted in order to analyze the specific variables connected to a good outcome of treatment for eating disorders (EDs; Fairburn, 2008a; Yager & Powers, 2007). However, much remains to be done to deepen the knowledge and increase treatment success of a category of diseases that, for its symptomatic heterogeneity, clinical complexity, diagnostic comorbidity, and resistance to change, is still today a real struggle for therapists (Vanderlinden, 2008; Westen & Harnden-Fischer, 2001). Among the

¹Department of Psychology, Milan Bicocca University, Italy. ²Department of Education Science and Technologies, "Tor Vergata" University of Rome, Italy.

^{Correspondence} concerning this paper should be addressed to Diego Sarracino PhD, Department of Psychology, Milan Bicocca University, Piazza dell'Ateneo Nuovo 1, 20126 Milano, Italy. E-mail: diego.sarracino@unimib.it therapeutic options available for EDs, CBT showed good efficacy, especially in the treatment of bulimia nervosa (National Institute for Health and Clinical Excellence, 2004; Shapiro et al., 2007; Wilson, Grilo, & Vitousek, 2007). In particular, Enhanced CBT (CBT-E; Byrne, Fursland, Allen, & Watson, 2011; Cooper & Fairburn, 2011; Fairburn, Cooper, & Shafran, 2008; Fairburn, Cooper, Shafran, Bohn, et al., 2008) and its variants (e.g., CBT-MS Multistep; Dalle Grave, 2003; Dalle Grave, Sartirana, Camporese, Marchi, & Calugi, 2007) are considered as the treatment of choice for EDs.

Despite these valuable therapeutic options, patients' dropout—i.e., a premature interruption of the treatment, due to a unilateral decision of the patient that is not negotiated with the therapist (Bergin & Garfield, 1994)—represents a serious challenge in ED treatments (Björk, Björck, Clinton, Sohlberg, & Norring, 2009; Blouin et al., 1995; Campbell, 2009; Sly, 2009). A review of the literature indicates that dropping out is quite common in both inpatient and outpatient treatments (Carter et al., 2012; Fairburn et al., 2009; Wallier et al., 2009). According to a recent metaanalysis, dropout rates from ED treatment range between 20 and 51% in inpatient settings, and 29 and 73% in outpatient settings (Fassino, Pierò, Tomba, & Abbate-Daga, 2009). It is important to study the factors that may influence the dropout from the therapy for several reasons (Carter et al., 2012; Fassino, Abbate-Daga, Pierò, Leombruni, & Rivera, 2003; Fassino et al., 2009). First, dropout patients are supposed to report worse prognosis and clinical outcomes (e.g., Baran, Weltzin, & Kaye, 1995). Second, dropping out is a very expensive occurrence in terms of direct and indirect costs for the mental health system (Mahon, 2000). Third, a high dropout rate decreases the power and generalizability of clinical trial results. Fourth, in the development of new therapeutic interventions, it is crucial to understand why individuals drop out in order to increase their engagement in the therapy and maximize treatment outcomes.

Despite the importance of this area of research, previous studies examining the factors affecting patient dropout from ED treatment have yielded fragmented and disparate results (Masson, Perlman, Ross, & Gates, 2007). Most dropout studies have focused on specific features of the disease itself (e.g., longer duration and severity of illness, presence of other comorbid disorders; Mahon, 2000) and a few dimensions regarding the patient's personality, such as high levels of impulsiveness and anger, high perfectionism, low self-esteem and self-directedness, strong avoidance and poor cooperativeness (Fairburn, Cooper, Shafran, Bohn, & Hawker, 2008; Fassino et al., 2003, 2009; Franzen, Backmund, & Gerlinghoff, 2004; Hoste, Zaitsoff, Hewell, & le Grange, 2007; Kahn & Pike, 2001; Peake, Limbert, & Whitehead, 2005; Steel et al., 2000). However, these varied and often contrasting findings do not clarify why many of these patients drop out of the therapy and how dropping out can be minimized. Recently, several authors have proposed to move from focusing exclusively on patient characteristics and adopt a broader approach that encompasses the whole process of treatment (Campbell, 2009; Carter, Pannekoek, Fursland, Allen, & Lampard, 2012; Sly, 2009). Kahn and Pike (2001) have also suggested that process-based factors, such as therapist-patient relationship and strength of the therapeutic alliance, may be more predictive of dropping out than individual patient characteristics (see also Carter et al., 2012).

The role of the patient-therapist relationship and therapeutic alliance in promoting psychotherapy change process and reducing the dropout rate is now well supported also for CBT (Gilbert & Leahy, 2007; Katzow & Safran, 2007; Leahy, 2008; Safran, 1998; Safran & Muran, 2000a; Waddington, 2002). This may be the case especially in difficult patients such as ED patients. One clinical trial on CBT for patients with anorexia nervosa (Ball & Mitchell, 2004) showed that the development of a strong therapeutic alliance was correlated with a better outcome. Fairburn, Cooper, and Shafran (2008) noticed a dropout reduction from 29% to 13% as a result of the modification of the CBT-E through a more extensive therapist openness towards dialogue, especially about the relationship with the patient. Fairburn and colleagues concluded that therapeutic alliance is also crucial for the importance that maintaining control has for ED patients. Also Piper and colleagues (1999) reported that a poor therapeutic alliance was associated with an increased dropout risk.

However, although there is some evidence that good therapeutic alliance in CBT for EDs is associated with better outcome and dropout reduction, less is known about how alliance develops and affects the course of this specific treatment. In their study comparing CBT and interpersonal therapy for bulimia nervosa, Constantino and Smith-Hansen (2008) suggested focusing on fluctuations and micro-fluctuations that facilitate or hamper the process of change during the whole therapy as well as each single session. The seminal work of Safran, Muran, and colleagues (reviewed by Safran & Muran, 2000a) has confirmed the existence of a rupture-repair pattern and its relation to session quality and treatment effectiveness (Katzow & Safran, 2007; Safran & Muran, 2000b; Strauss et al., 2006; see Colli & Lingiardi, 2009, for a review). Safran and Muran (1996) defined ruptures in terms of "deteriorations in the relationship between therapist and patient" (p. 447) that create a negative alteration in the quality of the therapeutic alliance. There is some evidence that in different therapeutic approaches, including CBT, lower rupture intensity is associated with a better interpersonal functioning, and that higher rupture resolution is connected to better outcomes and retention (Strauss et al., 2006; Waddington, 2002). In many studies, most ruptures resulted from therapists' rigid adherence to a therapeutic technique instead of an exploration of the client's difficult emotional experience and its impact (e.g., Ackerman & Hilsenroth, 2001).

Thus, even though there is some evidence that therapeutic alliance may be involved in outcome and dropping out, the precise manner in which the alliance influences the course of CBT treatment has not yet been explained in detail. The current study is the first to examine, session-by-session, some fundamental process variables—e.g., collaboration levels, ruptures, etc.—that can impair the achievement of a good outcome for ED treatment and, in some cases, lead to dropping out. We assumed that test-retest measurements alone are not sufficient to understand the factors that lead to dropping out, and that a thorough analysis of the changes and microchanges that occur during the therapeutic process, is essential in order to understand the evolution of the problem.

The present study used a single-case approach. Although single-case methodology is not necessarily the design of choice in all circumstances, it is a fundamental research strategy in psychotherapy process research—i.e., the study of the temporal unfolding of patient and patient-therapist variables within and across therapy sessions (Hilliard, 1993; Kazdin, 2010). Further, this approach may be necessary and/or preferable in pilot studies concerning difficult patients,

such as ED patients, for which, because of methodological issues and medical complications, it may be very problematic to have large samples or randomized clinical trials.

A purpose of this study is to present, along with standard test-retest assessment, observer ratings of patient and patient-therapist variables that may be hard to study by means of just self-reports. Moreover, we supposed that by assuming different perspectives (self-report questionnaires, therapist assessments and observer ratings) we would get a better understanding of the factors that lead to dropout.

Among the process variables studied, we focused on the therapeutic alliance and the ruptures in the therapeutic relationship that, if not repaired, may initiate and maintain a negative cycle in which the rupture remains unresolved. In addition, we examined in parallel the evolution of the patient's functioning in different areas (i.e., relationships, body, self-realization). These areas encompass aspects such as present and past family situation, quality of interpersonal relationships, presence of job-related problems, and onset of medical complications or other physical problems (for a review, see Mahon, 2000). In this regard, Sarracino and Dazzi (2007) identified sixteen areas to assess in detail at the beginning of the treatment and in subsequent sessions, in order to obtain a real-time representation of the experiences in patients' life (for a description of these areas, see Method section).

Specifically for the dropout case examined in the present study, our hypotheses were that (a) the patient's functioning would not improve during the treatment; (b) the therapeutic alliance global indices would not improve during the treatment; and (c) the patient-therapist collaboration levels and the rupture indices would show no improvement throughout the treatment.

Method

Case description

This is a single-case study of Emma, an Italian 22year-old woman, employee, who requested therapy because of her eating difficulties. From 3 to 13 years old, Emma lived in a boarding school, with only occasional contact with her biological parents. The biological father was an alcoholic who, when out of prison, lived as a tramp. During Emma's preadolescence, her natural parents split up and her mother moved to a different city. In this period, her visits to Emma were decreasingly frequent. When Emma was 13 years old, she was fostered by a couple. She perceived her foster parents as hypercritical, ambivalent and, in some cases, overtly unsupportive; however, in order to prevent the risk of being abandoned again, she responded to this new family situation with compliance and obedience. Her eating problems emerged for the first time in this period. In fact, her experience with her family showed a history of insecure attachment, in particular

with her mother, which may have a role in her internalizing and externalizing problems (e.g., Innamorati, Sarracino, & Dazzi, 2010; Sarracino, Presaghi, Degni, & Innamorati, 2011). The situation worsened further during the foster parents' separation, after which Emma chose to live with the foster father.

Before treatment, Emma received from the therapist an Axis-I diagnosis of bulimia nervosa, nonpurging type (DSM-IV-TR; American Psychiatric Association, 2000), characterized by the alternation of bulimic and anorexic phases. Her Axis-II diagnosis, according to the Structured Clinical Interview for DSM-IV Axis II (SCID-II; First, Gibbon, Spitzer, Williams, & Benjamin, 1997), was obsessive-compulsive personality disorder. SCID-II showed abnormal preoccupation with details and rules, exaggerated perfectionism, strict sense of duty in study and work, rigid moralism, and difficulties in interpersonal relationships. Despite these personality flaws, which affected especially the quality and quantity of her relationships, Emma was also very resourceful, and showed interest in dance, arts, cinema, literature, and politics.

Treatment description

Following the advice of a relative, Emma started an outpatient CBT conducted by a therapist with one year of clinical experience, in a specialized ED treatment center. The therapy was conducted on a weekly basis, 28 sessions overall, according to Dalle Grave's (2003) outpatient CBT-MS (multistep) model. Following Fairburn's transdiagnostic theory (Fairburn, 2008b; Fairburn, Cooper, & Shafran, 2003), this stepby-step approach aims at modifying the factors which underlie the whole ED spectrum. The treatment plan consists of three phases. The first phase (first six sessions) deals with problems regarding alimentation, binge-eating, compensatory behaviors, body weight, and other specific maintaining factors. The second phase addresses dysfunctional schemes of selfevaluation and possible additional maintaining factors (e.g., clinical perfectionism, low nuclear self-esteem, intolerance towards emotions, and interpersonal problems). This phase fills most of the time and its duration varies according to the BMI. More specifically, with BMI > 17.5, the therapy should be briefer (from 7 to 21 sessions), while with BMI < 17.5, the therapy requires more sessions (at least 45 sessions). The third phase aims to maintain the change achieved, prevent relapses, and clarify the opportunity of further therapies (last three bimonthly sessions). Consequently, given the low BMI of Emma (< 17.5; see Table 1), the therapist proposed a 48-session therapy clarifying this aspect to Emma as an essential component of the therapeutic contract in terms of positive improvement of her symptoms.

Emma initially agreed to this recommendation, but after 28 weekly sessions, she dropped out of the treatment. The therapist reaffirmed the importance of carrying on the therapy, because of the endurance of the eating symptomatology. However, Emma disagreed with the clinician and decided to interrupt the therapy because she was overwhelmed by family and financial problems.

After one year and three months, Emma returned to the therapist to "express gratitude for the work done" and update her about her decision to spend a period abroad, in order to distance herself from her family. In these last three sessions, the therapist encouraged the patient to reconsider her decision to drop out of therapy because of the persistence of serious eating problems (the patient reported both periods of restrictive diet and bulimic conduct) along with a significant withdrawal from social interactions. Regardless of this suggestion, Emma reaffirmed her strong intent to drop out permanently. At this point, the therapist asked Emma to keep her updated about future developments, but she never received any feedback.

Test-retest measures

Test-retest measures (0-6 months) were part of a standard assessment protocol of the clinical center and included standard instruments such as:

Symptom Checklist-90-Revised (SCL-90-R; Derogatis, 1994). It is a 90-item self-report instrument designed to measure current psychological and psychiatric symptoms. Each item is assessed on a Likert scale (from 0 to 4). Nine primary symptom dimensions are measured: Somatisation, Obsessive Compulsive, Interpersonal Sensitivity, Depression, Anxiety, Hostility, Phobic Anxiety, Paranoid Ideation, and Psychoticism. Also 3 global indices are obtained: the Global Severity Index (GSI), the mean Positive Symptom Total (PST) and the mean Positive Symptom Distress Index (PSDI). Most studies establish a T score > 62 for scales and global indices as a cut-off for clinical caseness. Overall, the instrument have shown good psychometric properties (Derogatis, 1994).

Eating Disorder Inventory (EDI-2; Garner, 1991). The EDI-2 is widely used in diagnosis of anorexia, bulimia and eating disorders not otherwise specified. It is made up of 11 scales assessing behavior, symptoms and personality traits of individuals with eating disorders: Drive for Thinness, Bulimia, Body Dissatisfaction, Ineffectiveness, Perfectionism, Interpersonal Distrust, Interceptive Awareness, Maturity Fears, Asceticism, Impulse Regulation, and Social Insecurity. A score higher than 14 in the Drive for Thinness scale is usually considered as a clinical cutoff point. The instrument showed good internal reliability (alpha ranging from .65 to .91 across scales in a clinical sample; Ebrenz & Gleavs, 1994), and a good testretest reliability (r ranging from .81 to .89 in an ED group; Thiel & Paul, 2006).

Toronto Alexithymia Scale (TAS-20; Bagby, Parker, & Taylor, 1994). The TAS is a 20-item instrument commonly used to measure alexithymia. Items are rated using a 5-point Likert scale whereby 1 = strongly disagree and 5 = strongly agree. The 20item version of the scale has 3 subscales: Difficulty Describing Feelings, Difficulty Identifying Feelings, and Externally-Oriented Thinking. The cut-off scores are: < 50 non-alexithymia; 51-59 possible alexithymia; > 60 alexithymia. The overall scale demonstrated good internal consistency (alpha = .81) and test-retest reliability (.77).

Sense of Belonging and Sharing Questionnaire (SBS; Procacci, Conversano, Semerari et al., 2002). The questionnaire consists of 90 items organized in a 7-point Likert format. It assesses the individual's feeling of belonging to a social group, and the belief to share values, objectives, interests, experiences etc. with the other members of the group. The assumption is that the sense of belonging is related to lower prevalence of metacognitive failures, social and emotional problems. Furthermore, two control scales (Lie and Frequency) are assessed. The questionnaire showed good internal reliability (alpha ranging from .68 to .93) and 40-day test-retest reliability (mean r = .70). Three indicative cutoffs are available: normal (T < 60), at risk (T = 61-64), and critical (T > 65).

Process measures

The process were evaluated including both the patient's functioning and the therapeutic alliance.

Motivational Areas Rating Scale (MARS; Sarracino & Dazzi, 2007). The MARS is an observer rating scale that provides a brief but wide-ranging research instrument for assessing patients' psychological distress from a trans-theoretical point of view. It consists of 16 macro-items, which cover the issues in four motivational areas—active relational (A), passive relational (B), corporeal (C), and self-realization (D), from four different perspectives—objective object (1), subjective object (2), objective self (3), and subjective self (4). For example, the A1 sub-area includes any act of hostility from others that is reported by the patient (e.g., physical or sexual assaults, offenses, injuries, threats, etc.), A2 includes problems in how patient subjectively experiences others (e.g., he/she feels anger, hatred, or suspicious towards others), A3 includes the patient's acts of hostility towards others, and so on (see Figure 2). These areas are defined motivational because they are related to the experiences in individuals' life that can motivate their behavior, affects, mental states, and contribute to their well-being and adjustment (for a complete description of motivational areas and their theoretical background, see Sarracino & Dazzi, 2007). The rater has to read the transcript regarding 2-session chunks and, based on the MARS manual, respond to the 16 items of the scale, assigning a score based on his or her perception of the salience and severity of the problem, from 0 (= absent) to 4 (= very high). In this way, we obtained a graph that illustrates Emma's functioning areas and their change over time (see Figure 2). In preliminary studies (Sarracino & Dazzi, 2007), MARS showed a good interrater reliability (ICCs from .52 to .81; p < .000).

Vanderbilt Therapeutic Alliance Scale - Short form (VTAS-R; Shelef & Diamond, 1998). This observer rating scale is composed of five items, which measure to what extent the patient: (1) Indicates that he/she experiences the therapist as understanding and supporting him/her; (2) Seems to identify with the therapist's method of working, so that he/she assumes part of the therapeutic task; and (3) Acts in a distrustful or defensive manner toward the therapist. Moreover, the scale assesses to what extent the therapist and patient together: (4) Share a common viewpoint about the definition, possible causes, and potential alleviation of patient's problems; and (5) Agree upon the goals and/or tasks for the session. The VTAS Short Form showed good psychometric properties, as regards validity and reliability, which are comparable to those of the 26-item version (Shelef & Diamond, 1998). Specifically for inter-rater reliability, ICCs ranged from .55 to .87. The scores of the 5 items, derived from chunks of two sessions, were summed to obtain a global scale distributed on 5 levels of increasing gravity: 0 = very high alliance (VTAS-R = 21-25);1 = high (16-20); 2 = medium (11-15); 3 = low (6-10);4 = very low(0-5).

Collaborative Interactions Scale (CIS; Colli & Lingiardi, 2009). This observer rating system is composed of two main scales: (a) The CIS-P, which assesses patient's direct rupture markers (DRMs; 9 items), indirect rupture markers (IRMs; 9 items), and collaborative processes (CPs; 3 items); and (b) The CIS-T, which assesses the therapist's positive interventions (PIs; 12 items) as well as negative interventions (NIs; 8 items). The rater assesses the presence, in each patient utterance, of a DRM, IRM, or CP and, in each therapist utterance, of a PI or NI. For each patient and therapist utterance, a rupture vs. collaboration score is also assessed from -3 (high rupture) to +3 (high collaboration). For each session, several global indices were obtained, each ranging from 0 to 30: index of direct ruptures (IDR), index of indirect ruptures (IIR), index of collaborative processes (ICP), index of negative interventions (INI), and index of positive interventions (IPI). CIS demonstrated good inter-rater reliability (average Cohen's k = .66-.81).

Procedure

The present study included three steps:

(1) A phase of psychodiagnostic assessment preceded the treatment and involved the administration to the patient of several self-reports (see *Test-retest measures* Section). Moreover, the patient reported (a) her current weight, (b) her greatest and lowest weight ever, and (c) her desired weight. After six months from the beginning of the treatment (after session 22, considering the summer holidays; see Figure 1), the same measures were retested. The Italian version was utilized for each self-report. Self-report questionnaires were part of the standard evaluation protocol of the clinical center, consequently they had been selected be-

fore the objectives of the present study were defined.

(2) The test-retest study was followed by a detailed analysis of the therapeutic process, based on a series of relevant therapeutic variables that were congruent with the aims of this study. Our aim was to tap possible divergences between the self-report results and the process analysis of the patient's functioning (MARS). Moreover, in this phase we used VTAS-R to achieve a quick screening of the level of therapeutic alliance reached during the treatment, in order to identify the most problematic moments of the therapy. In the light of the fact that MARS guidelines suggest clustering the sessions in 2-session chunks (Sarracino & Dazzi, 2007) we also decided to cluster VTAS-R scores in blocks of two, in order to facilitate comparisons between the two measures. Therefore, we obtaining a total of 14 codes (see Figure 2).

(3) Finally, we focused specifically on the therapeutic alliance between patient and therapist, in order to explore possible ruptures that might affect patient's decision to drop out of the treatment. For this purpose, CIS was used to analyze in detail sessions that resulted particularly critical and/or interesting in view of the therapeutic alliance through a preliminary analysis with the VTAS-R. Specifically, we focused on sessions 5, 18, and 23.

Process measures were coded separately by two graduating students in clinical psychology, who were trained and assisted in all phases of the assessment process by two expert researchers in this field. At the end of the assessment, any questions were discussed with the research assistant. The interrater reliability, in terms of intraclass correlation coefficients (Shrout & Fleiss, 1979) of the two judges for the three measures, was good to excellent (MARS: ICCs from .57 to .82; VTAS-R: ICCs from .60 to .81; CIS: average Cohen's *k* from .61 to .79).

Results

Test-retest comparisons

Table 1 shows Emma's height, weight, and T scores of self-report measures before treatment, and after six months. After 6 months, Emma reported a decrease of 15 kg in her current weight, to the extent that the cutoff score for underweight was reached. At the same time, the self-reported greatest weight increased, surpassing the overweight cutoff. The self-reported lowest weight however remained unchanged.

Before treatment, all SCL-90-R scales (except somtization) and global indices exceeded the cutoff point (T > 62). After 6 months, all scales were below the cutoff. Alexithymia scores (TAS-20) were below the cutoff both at test and retest assessments, and decreased substantially at the retest. At last, all SBS scales were below the cutoff, except the lie control scale— which reflects the tendency of the patient to give a positive, agreeable and socially desirable image of herself—which was high at both test and retest. Therefore

Table 1. Emma's height, weight, and T scores of self-report measures before treatment, and after six months

	-	Six- month
Measure	Test	retest
Height (cm)	171	_
Self-reported current weight in kg		
(BMI)	$66(23^{a})$	51 (17 ^b)
Self-reported greatest weight ever		
in kg (BMI)	69 (24ª)	75 (26°)
Self-reported lowest weight ever in		
kg (BMI)	48 (16 ^b)	48 (16 ^b)
Self-reported desired weight in kg		
(BMI)	57 (19ª)	55 (19ª)
SCL-90-R		
Somatization	59	49
Obsessive-compulsive	64 ^d	48
Interpersonal sensitivity	85 ^d	43
Depression	71 ^d	45
Anxiety	72 ^d	47
Hostility	76 ^d	47
Phobic anxiety	69 ^d	46
Paranoid ideation	73 ^d	46
Psychoticism	76^{d}	44
Global severity index (GSI)	76^{d}	45
Positive symptom total (PST)	64 ^d	46
Positive symptom distress Index		
(PSDI)	71^{d}	45
TAS-20		
Difficulty describing feelings	10	7
Difficulty identifying feelings	19	8
Externally-oriented thinking	17	15
Global alexithymia score	46	30
SBS		
Self-reflection	59	40
Representation of others' mind	49	39
Relational strategies	47	39
Painful feelings	63	47
Sense of superiority	40	41
Sense of detachment/indifference	55	50
Sense of belonging/sharing	51	40
Lie	66 ^d	66 ^d
Frequency	48	36

Note. BMI classification according to the WHO criteria: ^a Normal; ^b Overweight; ^c Underweight. ^d T scores higher than cutoff scores, according to the following criteria (see Method section for references): SCL-90-R: T > 62; TAS-20 global score: T > 60; SBS: T > 65.

re, the scores obtained by the patient at the SBS and other self-report measures, should be considered in the light of this bias.

Figure 1 shows Emma's EDI-2 profile before treatment, and after the first six months of treatment. This self-reported measure also shows a dramatic improvement in all the subscales. At the test, most scores were largely over the medium score obtained by the ED patients normative group. At the retest, only the perfectionism score exceeded the clinical cutoff. All the other self-reported measures were not only below the ED patients' normative group mean, but also below the female college normative group, resulting in a complete absence of symptoms for six scales.

Process analysis

Figure 2 shows a graphical, session-by-session representation of Emma's therapeutic process, including the evolution of the patient's functioning (MARS) and the therapeutic alliance (VTAS-R global index). The MARS indicates a moderate, pervasive presence of themes connected to real and perceived hostility from others (A1-2), which are particularly present in early and central sessions. The passive relational area (B) (i.e., issues of separation, withdrawal, and relational passivity) is very critical throughout the treatment, without remarkable improvements. The dysregulation of the corporeal area (C), as expected, is pervasive during all the therapy sessions, except for a brief moment (sessions 21-24). Working and existential problems (D1-3), and the resulting anxiety and insecurity, emerge at times during the treatment, principally in the latest sessions (when Emma decided to drop out of the treatment).

VTAS-R global index is low all through the treatment period, without substantial improvements.

Within-session examination of the therapeutic alliance

After the preliminary reading of the transcripts and the coding of VTAS-R global indices, we identified three sample sessions (5, 18, and 23) which seemed to be particularly interesting for the purpose of this study—i.e., which seemed to be representative of specific rupture styles (Table 2) or patterns of patienttherapist collaboration (Figure 3) in different stages of the treatment.

Session 5. This session was characterized by the absence of direct rupture markers from the patient, and by a low index of indirect ruptures. Therefore, the patient showed a great prevalence of collaborative processes of average intensity ($\mu = 1.56$). The therapist's positive interventions were quite consistent and varied as well. Nevertheless, the presence of a few negative interventions from the therapist is evident. Among negative interventions, the most frequent was NI5, indicating sudden changes of topic from the therapist, as in the following exchange:

- P: I realized that I idealized him [her ex-boyfriend] a lot [...]; it's clear I was wrong to think of him this way. [...]
- T: So you were feeling disappointed at that moment, and

Table 2. CIS scores for sessions 5, 18, and 23

CIS item		Presence (Intensity)		
		Session 18	Session 23	
DRM1: Patient does not agree with therapist about therapy tasks or goals	_	_	_	
DRM2: Patient criticizes therapist as a person or for his/her competence	-	-	-	
DRM3: Patient strongly refuses a therapist intervention or feels uncomfortable	-	1 (2)	7 (10)	
DRM4: Patient complains about lack of progress	-	-	_	
DRM5: Patient doubts about current session	-	-	-	
DRM6: Patient has doubts about being in therapy	-	-	-	
DRM7: Patient complains about parameters of therapy (e.g., session time, fee)	-	-	-	
DRM8: Patient has doubts about feeling better	-	_	_	
DRM9: Patient is sarcastic toward therapist	_	_	1(1)	
IRM1: Patient talks in wordy manner and/or talks about other people and their doings []	_	_	_	
IRM2: Patient changes topic or tangentially responds to therapist intervention	_	2 (2)	_	
IRM3: Patient gives short answer to therapist open question		3 (6)	2 (2)	
IRM4: Patient denies evident feeling state (e.g., anger, fear, shame)	1(1)	5 (9)	2 (3)	
IRM5: Patient intellectualizes his/her inner experience	1(1)	_	1(1)	
IRM6: Patient alludes to negative sentiments or concerns about therapeutic relationship	_	_	_	
IRM7: Patient interacts in a acquiescent manner	_	_	_	
IRM8: Patient uses self-enhancing strategies or self-justifying statements	_	_	1(1)	
IRM9: Patient is self-critical or self-blaming	_	_	_ (-)	
CP1: Patient talks about new significant fact, introduces a topic or elements within a topic	9 (14)	10 (12)	9 (13)	
CP2: Patient talks about his/her feeling and/or thoughts []	12 (18)	9 (11)	16 (23)	
CP3: Patient talks about meaning of events or connects topic to a topic or to a schema, etc.	3 (6)	5 (8)	2 (4)	
PI1: Therapist focuses on the here and now of the relationship	-	-	- (-)	
PI2: Therapist explores different patient states	4(5)	7 (9)	3 (3)	
PI3: Therapist explores different patient states	5 (8)	1 (2)	7 (9)	
PI4: Therapist suggests a patient emotion	5 (6)	4 (5)	4(7)	
PI5: Therapist believes that patient is indirectly talking about relationship	J (0)	I ())	-	
PI6: Therapist beneves that patient is indirectly taking about relationship PI6: Therapist furnishes empathic support to patient		2 (3)	5 (5)	
PI7: Therapist makes a clarification	4 (5)	2 (3) 4 (4)	8 (10)	
PI8: Therapist makes a confrontation	- (J)	- (T)	0(10)	
PI9: Therapist mates a comformation PI9: Therapist admits his/her participation in rupture process	_	_	_	
PI10: Therapist self-discloses countertransference feelings	-	1 (1)	-	
PI11: Therapist explains or redefines tasks/goals of therapy	-	1(1) 1(1)	-	
PI12: Therapist makes an interpretation	-		_	
NI1: Therapist makes an interpretation NI1: Therapist seems to impose his/her worldview or gives unwanted advice	-	- 1 (1)		
NI2: Therapist seems to compete with patient	-	1(1)	1(1)	
NI2: Therapist seems to compete with patient NI3: Therapist seems to press patient on specific topic	-	- 5 (6)	-	
	-	5 (6)	2 (3)	
NI4: Therapist seems doubtful about strategies	-	-	- (7)	
NI5: Therapist changes offhand topic	6(10)	1(1)	5(7)	
NI6: Therapist intellectualizes or is not focused on patient experience	1 (2)	3 (6)	1 (2)	
NI7: Therapist talks in technical jargon	-	-	-	
NI8: Therapist is hostile	-	-	-	
IDR	0	0.55	2.63	
IIR	0.74	4.72	1.67	
ICP	14.44	9.17	9.73	
IPI	10.34	8.59	9.50	
INI	4.43	3.89	3.08	

Note. CIS = Collaborative Interactions Scale; DRM = direct rupture marker; IDR = index of direct ruptures; IRM = direct rupture marker; IIR = index of indirect ruptures; CP = collaborative process; ICP = index of collaborative processes; PI = positive intervention; IPI = index of positive interventions; NI = negative intervention; INI = index of negative interventions. Global indexes (IDR, IIR, ICP, IPI, and INI) range from 0 to 30.

what did you do?

- P: I needed to tell someone I was disappointed, so I texted all my friends asking if I could call them [...] Then this morning a friend of mine called me, he was out yesterday when I texted him, and he asked me what had happened. And so in this difficult time in my life I shared something. I was happy to talk with someone [...] because I was home alone.
- T: And who was the first person you called from your friend list?

This type of patient-therapist interaction is found in many moments of the treatment. The patient often introduced topics of intense emotional impact, which the therapist did not explore appropriately. In fact, the therapist, despite staying on the same argument anyway, tended to shift the attention from the emotional field to more practical aspects.

As shown in Figure 3, the trend of the collaboration levels is broken with a series of ruptures, which are initially connected to therapists' interventions (point 13). Specifically, it is possible to see how, after a common rupture (17), a positive intervention from the therapist takes place. Despite not being initially captured (19), it provokes a collaborative intervention of the patient in the subsequent exchange (21, 22, 23). However, this intervention is not followed by an appropriate response from the therapist, who makes negative interventions.

Session 18. In this session, the quality of the therapeutic alliance considerably decreased compared to the previous phase of the treatment. One DRM emerged, and IRMs registered a substantial increase, with a prevalence of IRM4s, which refer to patient's mechanisms of negation of emotional states. Com-

pared to session 5, also the patient's collaboration level decreased, although Emma's attitude remained moderately collaborative. On the other hand, the therapist's positive and negative interventions remained almost unvaried, showing similar or slightly lower levels. Therapist's negative interventions nevertheless underwent a qualitative change: While initially focused on N15, they became more heterogeneous. This aspect can be interpreted as a worsening of the global alliance because negative interventions are not referred to a single area anymore, but begin to cover different fields. On several occasions the therapist's interventions assume in fact an inquiring rather than explorative attitude, as in the following example:

- T: And in your opinion, why does this happen?
- P: I don't know.
- T: Because he doesn't trust you?
- P: No, I don't think so. He said it is not for this reason.
- T: And so, how do you regard your father's attitude?
- P: Maybe he is testing me to see if my love is selfless. [...]
- T: But is it a topic that you spontaneously discuss?

It is clear that it is particularly difficult for both patient and therapist to maintain a collaborative communication in this session. Matching points between their collaboration processes are intermittent (3, 9-12, 15, 23, 24, 28, 31, 32, 36) and have short duration. Initially the patient kept up a moderately collaborative attitude, but the therapist's negative interventions discouraged the patient's efforts. In this way, the therapist caused a disruption (8) that was not repaired later in the session. In particular, in the following exchange regarding her foster father, there was a culminating moment where the opposite points of view of Emma



Figure 1. Emma's Eating Disorder Inventory-2 profile before treatment (test), and after six months (re-test). Pat = Eating disorder patient group (normative clinical sample); Nor = Female college group (normative non-clinical sample); Test = Emma's scores before treatment; Retest = Emma's scores at the six-month retest. DT = drive for thinness; B: bulimia; BD = body dissatisfaction; I = ineffectiveness; P = perfectionism; ID = interpersonal distrust; IA = interceptive awareness; MF = maturity fears; A = asceticism; IR = impulse regulation; SI = social insecurity.

and the therapist clashed irreparably:

- P: The less I know about his affairs the happier I am.
- T: But why don't you want to know? [...]
- P: Why should I know about all that? About his affairs?
- T: But why shouldn't you know? I mean, it is normal in a family to talk about these topics, isn't it?
- P: I don't know. It's all the same to me, it makes no difference.

Session 23. A crucial aspect concerning this session is the presence of a considerable number of DRM3s, which represent the situation when the patient firmly rejects an intervention or feels inconsolable:

T: You are undoubtedly a very strong girl also considering your age, you are mature. Nevertheless, even if you are and are seen to be strong by others [...] this doesn't mean that you have no needs, and so, in those moments where these needs are not satisfied, some vulnerable areas come out. Undoubtedly, [...] you don't want to fall into certain vicious circles again, and you are more confident about your personal resources [...]. You miss a sense of security from your family, but you know that nevertheless you are surrounded by love and warmth. Are you certain of this?

- P: Theoretically yes.
- T: Maybe not 100% certain.
- P: No.

The presence of DRM3 marker indicates that the patient does not take into account the therapist's point of view any longer, and maintains her conviction about the origins of her problems and possible solutions. In contrast with the increasing DRMs score, indirect rupture markers and negative interventions actually decreased (IIR = 1.67 vs. 4.72).

With regard to the collaboration levels, it proves more and more difficult for both therapist and patient to modulate and adapt their way of communicating. Collaborative exchanges decreased even further (5-6, 15-16, 30-35), indicating the difficulty for the patient to benefit from the interaction. Alliance ruptures came one after the other, from both patient and therapist, and were not repaired, as illustrated in the following exchange:



Figure 2. Session-by-session representation of Emma's motivational areas and therapeutic alliance. The graph illustrates the evolution of MARS areas and therapeutic alliance (VTAS-R global index), for 2-session blocks. The more problematic and unbalanced an area is, the darker it is. MARS levels (from white 0 to black 4): 0 = Problem not present or detectable; 1 = Recurrent but not intense problem, or intense but not recurrent problem; 3 = Very intense problem, or intense and recurrent problem; 4 = Intense and persistent problem that dominates patient's thoughts and behavior, or several intense problems belonging to the same category. Therapeutic alliance levels (from white 0 to black 4): 0 = very high alliance (VTAS-R = 21-25); 1 = high (16-20); 2 = medium (11-15); 3 = low (6-10); 4 = very low (0-5).

- P: So, you would like me to improve this relationship [with her foster father]
- T: At least you should try to change it
- P: I don't think it's possible
- T: So at the moment you think that going away is the best way [...]
- P: Yes, I do. [...] I would like to be more independent, I think our relationship would be better, I don't want to be constrained because...
- T: And what are you going to do about your mother?

This passage is particularly relevant: We observe how the therapist's interventions become progressively less adequate and synchronized. Now, patient and therapist move on two different levels of communication, with low chances of meeting one another.

Discussion

This study adopted a single-case, observer-rating approach in order to examine the role of process variables—particularly therapeutic alliance and alliance ruptures—in the dropout from outpatient CBT for EDs. As outlined previously, we assumed that, in the case discussed in this study, the patient's functioning and the therapeutic alliance would not improve during the treatment. Furthermore, we assumed that the patient-therapist collaboration levels and the rupture indices would show no improvement throughout the treat-

ment, indicating precipitating ruptures, mostly related to the therapist's failure to pay attention to tensions in the alliance or to the patient's subjective experience. Although further work is required to gain a more complete understanding of process-based factors involved in dropout from ED treatment, the results support our hypotheses, showing persistent impairment in multiple domains of the patient's life, and several serious ruptures in the patient-therapist relationship. These findings are in contrast with the fact that the patient reported impressive improvements in all self-reported measures. We may explain this discrepancy by hypothesizing that the patient wants to give a positive, agreeable and socially desirable image of herself (as suggested by the lie control scale), or in terms of the deep ambivalence towards the change that characterizes ED patients (Vitousek, Watson, & Wilson, 1998). A detailed examination of the patient's functioning areas during the treatment period shows a pervasive presence, at all stages of the treatment, of problematic areas concerning relationships, work and, obviously, her body. Issues of separation, hostility and withdrawal characterized the patient's functioning in her life and in her relationship with the therapist. The dysregulation of the corporeal area, as expected, was pervasive throughout the therapy. Anxiety and insecurity feelings also emerged, principally in the latest sessions, when the patient decided to drop out of the treatment.



Figure 3. Graphical representation of patient-therapist VTAS-R collaboration levels; *x*-axis: therapist-patient utterances; *y*-axis: -3 = high rupture; -2 = mid rupture; -1 = low rupture; +1 = low collaboration; +2 = mid collaboration; +3 = high collaboration.

Our case study also supported the importance of patient-therapist relationship and therapeutic alliance in affecting the patient's engagement in the treatment. Overall, the patient-therapist collaboration was broken by a series of ruptures, giving the impression that their interaction became difficult, that comprehension and deep confidence that enable an effective communication were withering. Alliance ruptures came one after the other, from both patient and therapist, and were not repaired. In particular, the rising of DRMs in the later sessions can be interpreted as the result of a progressive fall in the patient's confidence in her possibility to receive concrete help from the therapist. The verbal exchanges where the patient no longer took take the therapist's point of view into account were increasingly frequent. In fact, apparently the patient maintained her conviction about the origins of her problems and possible solutions. By examining these late sessions, there are clear signs of the forthcoming dropout. However, the therapist was not able to realize this until the very moment when the patient decided unilaterally to interrupt the treatment prematurely, and therefore she just had to accept the fait accompli. This is consistent with Bergin and Garfield's (1994) definition of drop out as a premature interruption of the treatment, due to a unilateral decision of the patient that is not negotiated with the therapist. Although, after one year and three months, Emma returned to the therapist to "express gratitude for the work done," we should consider this fact in the light of the patient's high scores on the SBS "lie scale," which may be an indicator of social desirability (Procacci et al., 2002).

This difficulty of communication between patient and therapist is illustrated by the fact that every time the patient introduced topics of intense emotional impact, the therapist failed to explore these feelings more deeply, shifting the focus from the emotional field to concrete matters. This negative intervention recurred throughout the treatment, and did not allow the patient to value her own experience, and to know that her therapist knew how important this was for her. In line with previous research (e.g., Ackerman & Hilsenroth, 2001; Castonguay, Goldfried, Wiser, Raue, & Hughes, 1996), such ruptures may depend on the tendency for therapists to respond to strains in the relationship by persisting dogmatically with the application of a therapeutic technique rather than exploring the patient's difficult emotional experience and its impact. The risk that the therapeutic alliance may be overshadowed by techniques and protocols (Mahoney, 1991) is more pronounced for inexperienced clinicians, as in our case; however, the difficulty in managing one's own involvement in interpersonal conflict may be also present in skilled and highlytrained clinicians (Binder & Strupp, 1997).

Therefore, our findings support the argument outlined by Leahy (2008) that the quality of the relationship and the patient's perception of the therapeutic alliance are related to better outcomes and reduced dropout rates, and therefore should be considered as key factors in CBT at least as much as cognitive therapists' skills. Patients and clinicians should work together aiming at allowing the patient to become an active participant in the understanding of what is going on in the treatment. Failure to recognize rupture markers may result in the tendency for therapists to unintentionally exacerbate patients' distress or disappointment and not address significant problems; this may lead to premature treatment termination. On the contrary, resolving ruptures in the therapeutic relationship provides an often essential opportunity for using the relationship as a mean to modify cognitive and emotional problems (Aspland, Llewelyn, Hardy, Barkham, & Stiles, 2008; Katzow & Safran, 2007).

A first strength of this study is to explore, for the first time, *how* therapeutic alliance affects the course of CBT for EDs. We adopted an approach encompassing the whole therapy process; this allowed us to examine session-by-session the evolution of some fundamental changes and microchanges in the therapist-patient relationship and in the patient's functioning, which may affect his or her attitude toward the treatment. Our findings are valuable because processbased factors, such as rupture markers or collaboration levels, are arguably easier to address and modify than individual patient characteristics (Carter et al., 2012). It would be desirable for therapists to have a real-time assessment of these variables during their therapies, in order to flexibly modify their interventions, repair ruptures, and prevent dropout; however, in clinical practice, there is still a long way to go. Second, this study has the merit of using, along with standard test-retest self-reported assessment, independent observer ratings, so as to assume different perspectives on the therapeutic process. The importance of integrating patient and therapist measures with observer ratings is supported by a meta-analysis of 24 clinical studies including different therapies, showing that the effect size of alliance predicting outcome varied depending on the source of outcome report (i.e., client, therapist, or observer; Horvath & Symonds, 1991). It should be noted, moreover, that self-reports should be used cautiously with patients suffering from more serious disorders, because of the risk of biases and manipulation (e.g., patients who minimize or exaggerate their progresses to bring therapist's attention to their illness or to benefit from "secondary advantages"; Corcoran & Fischer, 2000). Therefore, our study emphasizes the usefulness of an integrated use of different process methodologies and perspectives.

The generalizability of this study is limited for several reasons. First, we adopted a single-case approach, and future research is needed to replicate these findings with different therapists, patients, and ED disorders. Second, previous research has indicated that process factors other than those identified in this study are likely to play a role in predicting treatment dropout. For example, the patient's "readiness to change" may have an important role in dropout from ED treatment (Bewell & Carter, 2008). Other variables of interest are the emotional reactions and cognitions that both patients and psychotherapists experience towards each other; it would be interesting to examine these variables not only in the psychodynamic treatments focused on transference and countertransference, but also in CBT, especially for difficult patients (Prasko et al., 2010). Further research is required to identify the whole range of process variables that may influence patient's dropout from the treatment. A third limitation of this study is that the same two raters evaluated all the sessions with all the measures, creating a bias about the possible "contamination" of ratings. In order to solve this problem, future studies might also include a process analysis based on computerized methods (e.g., Lo Verde, Sarracino, & Vigorelli, 2012). Finally, we did not apply any measures of therapist treatment adherence: therefore, it is difficult to say if the therapist really provided this intervention in its correct form.

In conclusion, one strength of this study is to consider the "therapeutic relationship or alliance as an ongoing process, rather than an achievement that is fixed at one point in time, since the relationship is interactive and iterative, reflecting the patient's response to the therapist's response to the patient" (Leahy, 2008, p. 770). This approach combines a clinical interest to respond appropriately to changes within the patient, and a research interest to find support for the role of these changes in the patient's decision to drop out of the treatment.

Acknowledgements

We would like to thank Antonino Carcione, Marina Russo and Stefania D'Angerio for their support in the early stages of the study.

References

- Ackerman, S. J., & Hilsenroth, M. J. (2001). A review of therapist characteristics and techniques negatively impacting the therapeutic alliance. *Psychotherapy*, 38(2), 171–185. doi: 10.1037/0033-3204.38.2.171
- American Psychiatric Association (2000). *Diagnostic and statistical manual of mental disorders* (4th ed., text revision). Washington, DC: American Psychiatric Association.
- Aspland, H., Llewelyn, S., Hardy, G. E., Barkham, M., & Stiles, W. (2008). Alliance ruptures and rupture resolution in cognitivebehavior therapy: A preliminary task analysis. *Psychotherapy Research*, 18(6), 699–710. doi: 10.1080/10503300802291463
- Bagby, R. M., Parker, J. D. A., & Taylor, G. J. (1994). The twentyitem Toronto Alexithymia Scale-II. Item selection and crossvalidation of the factor structure. *Journal of Psychosomatic Research*, 38(1), 23–32. doi: 10.1016/0022-3999(94)90006-X
- Ball, J., & Mitchell, P. (2004). A randomized controlled study of cognitive behavior therapy and behavioral family therapy for anorexia nervosa patients. *Eating Disorders: The Journal of Treatment & Prevention*, 12(4), 303–314. doi: 10.1080/10640260490521389
- Baran, S. A., Weltzin, T. E., & Kaye, W. H. (1995). Low discharge weight and outcome in anorexia nervosa. *American*

Journal of Psychiatry, 152(7), 1070-1072.

- Bergin, A. E., & Garfield, S. S. (1994). *Handbook of psychotherapy and behavior change*. New York: Wiley.
- Bewell, M. A., & Carter, J. C. (2008). Readiness to change mediates the impact of eating disorder symptomatology on treatment outcome in anorexia nervosa. *International Journal of Eating Disorders*, 41(4), 368–371. doi: 10.1002/eat.20513
- Binder, J. L., & Strupp, H. H. (1997). 'Negative process': A recurrently discovered and underestimated facet of therapeutic process and outcome in the individual psychotherapy of adults. *Clinical Psychology: Science and Practice*, 4(2), 121– 139. doi: 10.1111/j.1468-2850.1997.tb00105.x
- Björk, T., Björck, C., Clinton, D., Sohlberg, S., & Norring C. (2009). What happened to the ones who dropped out? Outcome in eating disorder patients who complete or prematurely terminate treatment. *European Eating Disorders Review*, 17, 109–119.
- Blouin, J., Schnarre, K., Carter, J., Blouin, A., Tener, L., Zuro, C., & Barlow, J. (1995). Factors affecting dropout rate from cognitive-behavioural group treatment for bulimia nervosa. *International Journal of Eating Disorders*, 17(4), 323–329. doi: 10.1002/erv.911
- Byrne, S. M., Fursland, A., Allen, K. L., & Watson, H. (2011). The effectiveness of enhanced cognitive behavioural therapy for eating disorders: An open trial. *Behaviour Research and Therapy*, 49(4), 219–226. doi: 10.1016/j.brat.2011.01.006
- Campbell, M. (2009). Drop-out from treatment for eating disorders: A problem for clinicians and researchers. *European Eating Disorders Review*, 17(4), 239–242. doi: 10.1002/erv.934
- Carter, O., Pannekoek, L., Fursland, A., Allen, K. L., & Lampard, A. M. (2012). Increased wait-list time predicts dropout from outpatient enhanced cognitive behaviour therapy (CBT-E) for eating disorders. *Behaviour Research and Therapy*, 50, 487–492. doi: 10.1016/j.brat.2012.03.003
- Castonguay, L. G., Goldfried, M. R., Wiser, S., Raue, P. J., & Hughes, A. M. (1996). Predicting the effects of cognitive therapy for depression: A study of unique and common factors. *Journal of Consulting and Clinical Psychology*, 63(7-8), 497–504.
- Colli, A., & Lingiardi, V. (2009). The Collaborative Interactions Scale: A new transcript-based method for the assessment of therapeutic alliance ruptures and resolutions in psychotherapy. *Psychotherapy Research*, 19(6), 718–734. doi: 10.1080/10503300903121098
- Constantino, M., & Smith-Hansen, L. (2008). Patient interpersonal factors and the therapeutic alliance in two treatments for bulimia nervosa. *Psychotherapy Research*, 18(6), 683–698. doi: 10.1080/10503300802183702
- Cooper, Z., & Fairburn, C. G. (2011). The evolution of 'enhanced' cognitive behavior therapy for eating disorders: Learning from treatment nonresponse. *Cognitive and Behavioral Practice*, 18(3), 394–402. doi: 10.1016/j.cbpra.2010.07.007
- Corcoran, K., & Fischer, J. (2000). *Measures for clinical practice*. *A sourcebook* (3rd ed., vol. 1). New York: Free Press.
- Dalle Grave, R. (2003). Terapia cognitivo-comportamentale ambulatoriale dei disturbi dell'alimentazione [Outpatient cognitive behavioural therapy for eating disorders]. Verona: Positive Press.
- Dalle Grave, R., Sartirana, M., Camporese, L., Marchi, I., & Calugi, S. (2007). Terapia cognitivo-comportamentale multistep per i disturbi dell'alimentazione: Basi teoriche e aspetti pratici [Multistep cognitive behavioural therapy for eating disorders: Theoretical basis and practical aspects]. *Cognitivismo Clinico*, 4(1), 1–18.
- Derogatis, L.R. (1994). Symptom Checklist-90-R: Administration, scoring, and procedures manual (3rd ed.). Minneapolis, MN: National Computer Systems.
- Fairburn, C. G. (Ed.) (2008a). *Cognitive behavior therapy and eating disorders*. New York: Guilford.
- Fairburn, C. G. (2008b). Eating disorders: The transdiagnostic

view and the cognitive behavioral theory. In C. G. Fairburn (Ed.), *Cognitive behavior therapy and eating disorders* (pp. 7–22). New York: Guilford.

- Fairburn, C. G., Cooper, Z., Doll, H. A., O'Connor, M. E., Bohn, K., Hawker, D. M., ... Palmer, R. L. (2009). Transdiagnostic cognitive-behavioral therapy for patients with eating disorders: A two-site trial with 60-week follow-up. *American Journal of Psychiatry*, 166(3), 311–319. doi: 10.1176/appi.ajp.2008.08040608
- Fairburn, C. G., Cooper, Z., & Shafran, R. (2003). Cognitive behavior therapy for eating disorders: A "transdiagnostic" theory and treatment. *Behavior Research and Therapy*, 41(5), 509–528. doi: 10.1016/S0005-7967(02)00088-8
- Fairburn, C. G, Cooper, Z., & Shafran, R. (2008). Enhanced cognitive behavior therapy for eating disorders (CBT-E): An overview. In C. G. Fairburn (Ed.), *Cognitive behavior therapy* and eating disorders (pp. 23–34). New York: Guilford.
- Fairburn, C. G., Cooper, Z., Shafran, R., Bohn, K., & Hawker, D. M. (2008). Clinical perfectionism, core low self-esteem and interpersonal problems. In C. G. Fairburn (Ed.), *Cognitive behavior therapy and eating disorders* (pp. 197–220). New York: Guilford.
- Fairburn, C. G., Cooper, Z., Shafran, R., Bohn, K., Hawker, D. M., Murphy, R., & Straebler, S. (2008). Enhanced cognitive behavior therapy for eating disorders: The core protocol. In C. G. Fairburn (Ed.), *Cognitive behavior therapy and eating disorders* (pp. 45–193). New York: Guilford.
- Fassino, S., Abbate-Daga, G., Pierò, A., Leombruni, P., & Rovera G. G. (2003). Dropout from brief psychotherapy within a combination treatment in bulimia nervosa: The role of temperament and anger. *Psychotherapy and Psychosomatics*, 72(4), 203–210. doi: 10.1159/000070784
- Fassino, S., Pierò, A., Tomba, E., & Abbate-Daga, G. (2009). Factors associated with dropout from treatment for eating disorders: A comprehensive literature review. *BMC Psychiatry*, 9, 67–75. doi: 10.1186/1471-244X-9-67
- First, M. B., Gibbon, M., Spitzer, R. L., Williams, J. B. W., & Benjamin, L. S. (1997). Structured Clinical Interview for DSM-IV disorders Axis II (SCID-II). Washington, DC: American Psychiatric Press.
- Franzen, U., Backmund, H., & Gerlinghoff, M. (2004). Day treatment group programme for eating disorders: reasons for drop-out. *European Eating Disorders Review*, 12(4), 153– 158. doi: 10.1002/erv.515
- Garner, D. M. (1991). *Eating Disorder Inventory-2*. Odessa, FL: Psychological Assessment Resources.
- Gilbert, P., & Leahy, R. L. (Eds.). (2007). The Therapeutic Relationship in the Cognitive Behavioral Psychotherapies. New York: Routledge.
- Hilliard, R. B. (1993). Single-case methodology in psychotherapy process and outcome research. *Journal of Consulting and Clinical Psychology*, 61(3), 373–380. doi: 10.1037/0022-006X.61.3.373
- Horvath, A. O., & Symonds, B. D. (1991). Relation between working alliance and outcome in psychotherapy: A metaanalysis. *Journal of Counseling Psychology*, 38(2), 139–149. doi: 10.1037/0022-0167.38.2.139
- Hoste, R. R., Zaitsoff, S., Hewell, K., & le Grange, D. (2007). What can dropouts teach us about retention in eating disorder treatment studies? *International Journal of Eating Disorders*, 40(7), 668–671. doi: 10.1002/eat.20421
- Innamorati, M., Sarracino, D., & Dazzi, N. (2010). Motherhood constellation and representational change in pregnancy. *Infant Mental Health Journal*, 31(4), 379–396, doi: 10.1002/imhj.20261
- Kahn, C., & Pike, K. M. (2001). In search of predictors of dropout from inpatient treatment for anorexia nervosa. *International Journal of Eating Disorders*, 30(3), 237–244. doi: 10.1002/eat.1080

- Katzow, A. W., & Safran, J. D. (2007). Recognizing and resolving ruptures in the therapeutic alliance. In P. Gilbert & R. L. Leahy. (Eds.), *The Therapeutic Relationship in the Cognitive-Behavioural Psychotherapies* (pp. 90–105). New York: Routledge.
- Kazdin, A. E. (2010). Single-case research designs: Methods for clinical and applied settings (2nd ed.). New York: Oxford University Press.
- Lo Verde, R., Sarracino, D., & Vigorelli, M. (2012). Therapeutic cycles and referential activity in the analysis of the therapeutic process. *Research in Psychotherapy: Psychopathology, Process and Outcome*, *15*(1), 22–31, doi: 10.7411/RP.2012.003
- Leahy, R. L. (2008). The therapeutic relationship in cognitivebehavioral therapy. *Behavioural and Cognitive Psychotherapy*, 36(6), 769–777. doi: 10.1017/S1352465808004852
- Mahon, J. (2000). Dropping out from psychological treatment for eating disorders: What are the issues? *European Eating Disorders Review*, 8(3), 198–216.
- Mahoney, M. J. (1991). Human Change Processes: The Scientific Foundations of Psychotherapy. New York: Basic Books.
- Masson, P. C., Perlman, C. M., Ross, S. A., & Gates, A. L. (2007). Premature termination of treatment in an inpatient eating disorder programme. *European Eating Disorders Review*, 15, 275–282. doi: 10.1002/(SICI)1099-0968(200005) 8:3<198:AID-ERV356>3.0.CO;2-3
- National Institute for Health and Clinical Excellence. (2004). Core interventions in the treatment and management of anorexia nervosa, bulimia nervosa and related disorders. Retrieved from http://www.nice.org.uk/CG009
- Peake, K. J., Limbert, C., & Whitehead, L. (2005). Gone but not forgotten: an examination of the factors associated with dropping out from treatment of eating disorders. *European Eating Disorders Review*, 13(5), 330–337. doi: 10.1002/erv.645
- Piper, W., Ogrodniczuk, J., Joyce, A., McCallum, M., Rosie, J., O'Kelly, J., & Steinberg, P. (1999). Prediction of dropping out in time-limited, interpretive individual psychotherapy. *Psychotherapy: Theory, Research, Practice, Training*, 36(2), 114–122. doi: 10.1037/h0087787
- Prasko, J., Diveky, T., Grambal, A., Kamaradova, D., Mozny, P., Sigmundova, Z., Slepecky, M., & Vyskocilova, J. (2010). Transference and countertransference in cognitive behavioral therapy. *Biomedical Papers*, 154(3), 189–198. doi: 10.5507/bp.2012.027
- Procacci, M., Conversano, P., Semerari, A., Nicolò, G., Cordone, A., Dimaggio, G., Candilera, G. & Iliceto, P. (2002). Il questionario sul senso di appartenenza e condivisione (SAC): Costruzione e validazione dello strumento [Sense of belonging and sharing questionnaire: Construction and validation of the instrument]. *Ricerca in Psicoterapia*, 5(3), 123–140.
- Safran, J. D. (1998). Widening the Scope of Cognitive Therapy: The Therapeutic Relationship, Emotion and the Process of Change. Northvale, NJ: Aronson.
- Safran, J. D., & Muran, J. C. (1996). The resolution of ruptures in the therapeutic alliance. *Journal of Consulting and Clinical Psychology*, 64(3), 447–458. doi: 10.1037/0022-006X.64.3.447
- Safran, J. D., & Muran, J. C. (2000a). Negotiating the Therapeutic Alliance: A Relational Treatment Guide. New York: Guilford.
- Safran, J. D., & Muran, J. C. (2000b). Resolving therapeutic alliance ruptures: Diversity and integration. *Journal of Clini*cal Psychology, 56(2), 233–243. doi: 10.1002/(SICI)1097-4679(200002)56:2<233::AID-JCLP9>3.0.CO;2-3
- Sarracino, D., & Dazzi, N. (2007). Motivational Areas Rating Scale: Developing a new tool for studying psychotherapy process and outcome. *Psychotherapy Research*, 17(6), 721– 732. doi: 10.1080/10503300701303369
- Sarracino, D., Presaghi, F., Degni, S., & Innamorati, M. (2011). Sex-

specific relationships among attachment security, social values, and sensation seeking in early adolescence: implications for adolescents' externalizing problem behaviour. *Journal of Adolescence*, *34*(3), 541–554. doi: 10.1016/j.adolescence.2010.05.013

- Shapiro, J., Berkman, N. D., Brownley, K. A., Sedway, J. A., Lohr, K. N., & Bulik, C. M. (2007). Bulimia nervosa treatment: A systematic review of randomised controlled trials. *International Journal of Eating Disorders*, 40(4), 321–336. doi: 10.1002/eat.20372
- Shelef, K., & Diamond, G. M. (2008). Short form of the revised Vanderbilt Therapeutic Alliance Scale: Development, reliability, and validity. *Psychotherapy Research*, 18(4), 433–443. doi: 10.1080/10503300701810801
- Shrout, P. E., & Fleiss, J. L. (1979). Intraclass correlations: Uses in assessing rater reliability. *Psychological Bulletin*, 86(2), 420–428. doi: 10.1037/0033-2909.86.2.420
- Sly, R. (2009). What's in a name? Classifying 'the dropout' from treatment for anorexia nervosa. *European Eating Disorders Review*, 17(6), 405–407. doi: 10.1002/erv.964
- Steel, Z., Jones, J., Adcock, A., Clancy, R., Bridgford-West, L., & Austin, J. (2000). Why the high rate of dropout from individualized cognitive-behaviour therapy for bulimia nervosa? *International Journal of Eating Disorders*, 28(2), 209– 214. doi: 10.1002/1098-108X(200009)28:2<209::AID-EAT10>3.0.CO;2-L
- Strauss, J. L., Hayes, J., Newman, C. F., Brown, G. K., Barber, J. P., Laurenceau, J. P., & Beck, A. T. (2006). Early alliance, alliance ruptures, and symptom change in nonrandomized trial of cognitive therapy for avoidant and obsessive-compulsive personality disorders. *Journal of Consulting and Clinical Psycholo*gy, 74(2), 337–345. doi: 10.1037/0022-006X.74.2.337

Thiel, A., & Paul, T. (2006). Test-retest reliability of the Eating

Disorder Inventory 2. Journal of Psychosomatic Research, 61(4), 567-569. doi: 10.1016/j.jpsychores.2006.02.015

- Vanderlinden, J. (2008). Many roads lead to Rome: Why does cognitive behavioural therapy remain unsuccessful for many eating disorder patients? *European Eating Disorders Review*, 16(5), 329–333. doi: 10.1002/erv.889
- Vitousek, K., Watson, S., & Wilson, G. (1998). Enhancing motivation for change in treatment-resistant eating disorders. *Clinical Psychology Review*, 18(4), 391–420. doi: 10.1016/S0272-7358(98)00012-9
- Waddington, L. (2002). The therapy relationship in cognitive therapy: A review. Behavioral and Cognitive Psychotherapy, 30(2), 179–191. doi: 10.1017/S1352465802002059
- Wallier, J., Vibert, S., Berthoz, S., Huas, C., Hubert, T., & Godart, N. (2009). Dropout from inpatient treatment for anorexia nervosa: Critical review of the literature. *International Journal of Eating Disorders*, 42(7), 636–647. doi: 10.1002/eat.20609
- Westen, D., & Harnden-Fischer, J. (2001). Personality profiles in eating disorders: Rethinking the distinction between axis I and axis II. *American Journal of Psychiatry*, 158(4), 547–562. doi: 10.1176/appi.ajp.158.4.547
- Wilson, G. T., Grilo, C. M., & Vitousek, K. M. (2007). Psychological treatments of eating disorders. *American Psychologist*, 62(3), 199–216. doi: 10.1037/0003-066X.62.3.199
- Yager, J., & Powers, P. S. (Eds.). (2007). Clinical Manual of Eating Disorders. Washington, DC: American Psychiatric Publishing.

Received April 30, 2013 Revision June 7, 2013 Accepted October 2, 2013